**Author's response to reviews**

**Title:** Association between intraarticular cytokine levels and clinical parameters of osteochondritis dissecans in the ankle

**Authors:**

Hagen Schmal (hagen.schmal@freenet.de)
Ingo H Pilz (ingo.hinrich.pilz@uniklinik-freiburg.de)
Ralf Henkelmann (ralfhenkelmann@googlemail.com)
Gian M Salzmann (giansalzmann@yahoo.com)
Norbert P Südkamp (norbert.suedkamp@uniklinik-freiburg.de)
Philipp Niemeyer (PhNiemeyer@gmail.com)

**Version:** 4  
**Date:** 12 April 2014

**Author's response to reviews:**

Reviewer's comments:

1. Please add the Trial Registration Number date as the last line of the abstract next to the Trial Registration Number.

   This was added.

2. Please move the Competing interest section and Author's contribution section before the Acknowledgement section. Please note of the following:

   - Title page
   - Abstract
   - Keywords
   - Background
   - Methods
   - Results and discussion
   - Conclusions
   - List of abbreviations used (if any)
   - Competing interests
   - Authors' contributions
   - Authors' information
   - Acknowledgements
   - Endnotes
   - References
   - Illustrations and figures (if any)
Tables and captions
This was changed.